Author's response to reviews

Title: Melasma and its association with different types of nevi in women: A case control study

Authors:

Hassan Adalatkha (h.adalatkhah@yahoo.com)
Homayoun Sadeghi-Bazargani (homsad@ki.se)
Nayereh Aminisani (n.aminisanibeja@yahoo.com)
Zeynizade Somayyeh (Somayyeh.zeynizade@yahoo.com)

Version: 6 Date: 11 June 2008

Author's response to reviews: see over
Dear Editor
Thanks for final comments of the reviewers, I revised the manuscript based on two comments given by the respected reviewer. The point by point responses are as follows:

Reviewer1:
No further revision requested.

Reviewer2:
1- The statement regarding etiologic similarities between melasma and nevi was deleted as recommended by the reviewer.
2- Language edition / Done

Best regards
Corresponding author
Homayoun Sadeghi MD, MPH
PHS department, Karolinska Institute
2nd floor, Norrbacka
SE-171 76
Stockholm, Sweden